Author's response to reviews

Title: Solitary Osteochondroma of the Twelfth Rib with Intraspinal Extension and Cord Compression in Middle-Aged Patient

Authors:

Jung Hyun Shim (gumjung70@gmail.com)
Choon Keun Park (allspine@gmail.com)
Seung Ho Shin (sinsho7@gmail.com)
Hee Sun Jeong (radhsjung@naver.com)
Jang Hoe Hwang (jhhwang@hallym.ac.kr)

Version: 2 Date: 19 July 2011

Author's response to reviews:

we added "Consent" in the Acknowledgements section, as follows: "Written informed consent was obtained from the patient for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."